Author’s response to reviews

Title: Hypertension Outcomes of Adrenalectomy in Patients with Primary Aldosteronism: a Systematic Review and Meta-analysis

Authors:

Zhou Yu (510713651@qq.com)
Zhang Meilian (1024295955@qq.com)
Ke Sujie (1924210179@qq.com)
Liu Libin (Libin.Liu@hotmail.com)

Version: 3 Date: 09 Sep 2017

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you again for your letter and comments concerning our manuscript entitled “Hypertension Outcomes of Adrenalectomy in Patients with Primary Aldosteronism: a Systematic Review and Meta-analysis” (ID: BEND-D-16-00217). The main corrections in the paper and the responds to the comments are as following:

1. Response to comment: There is still a discrepancy between the email addresses in Editorial Manager and your title page. These must be the same. Please amend accordingly. If you cannot access one of the email addresses any more than please remove it.

Response: We have amend the email addresses in Editorial Manager in accordance with our title page.

2. Response to comment: We have noted that the names of the authors are in a different configuration in Editorial Manager compared to the title page, e.g. Yu Zhou and Zhou Yu. Please
ensure these are listed the same in Editorial Manager and the title page, and amend the initials in the Authors' Contributions accordingly.

Response: We have amend the names of the authors in Editorial Manager in accordance with our title page.

3. Response to comment: Please change the heading Ethics statement to Ethics approval and consent to participate.

Response: We have changed the heading Ethics statement to Ethics approval and consent to participate in our manuscript.

4. Response to comment: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We will upload our files in accordance with above demand.

Once again, we appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Corresponding author:

Name: Libin liu

E-mail: Libin.Liu@hotmail.com